# INVESTIGATION OF CETUXIMAB RESISTANCE IN HEAD AND NECK SQUAMOUS CELL CARCINOMAS

# FINAL REPORT ID: K116295 PI: József Tóvári, PhD

#### **1 BACKGROUND:**

Squamous cell carcinomas of the Head and Neck region (HNSCCs) are the sixth most abundant malignancies, responsible for over 450.000 deaths annually <sup>1</sup>. Patients with recurrent/metastatic tumors receive first line combination chemotherapy as palliative treatment. This, according to the EXTREME protocol, involves six cycles of cisplatin/carboplatin and 5-fluorouracil treatment, alongside with the targeted treatment against the Epidermal Growth Factor Receptor (EGFR): cetuximab <sup>2</sup>. After six cycles, cetuximab monotherapy continues until progression. This combination regimen proved to be superior to monotherapies, however, response rates are low, and resistance often leads to early progression. Therefore, understanding (and avoiding) the mechanism of resistance is an urgent need. Among other factors (tumor hypoxia, tumor immunoevasion, tumor microbiome), the extracellular alterations of the transmembrane receptor EGFR were reportedly abundant in HNSCC and were suspected to contribute to therapy failure <sup>3</sup>. Also, alternative pathways of cellular survival and proliferation signaling, such as PI3K-AKT-mTOR pathway, or c-MET receptor based signaling was suspected to be involved <sup>4</sup>. However, the results published were unclear whether the changes in the receptor or alternative signaling might be important in cetuximab-resistance. Clarification of the problem might be of vital importance for many patients.

#### 2 **RESULTS:**

#### 2.1 EGFR, and cetuximab resistance in HNSCC

The problem implied the accurate screening of EGFR alterations in experimental systems and in clinical HNSCC samples as well. We found that, with huge variance among published data, the multi-exon deletion resulting truncated receptor variant (EGFR vIII), and the single nucleotide missense polymorphism (EGFR R521K) were reported to be abundant in HNSCC patients. Therefore, we performed targeted PCR of the crucial areas. For EGFR vIII detection, we used primer sequences for EGFR vIII: 5'-GCT CTG GAG GAA AAG AAA GGT AAT TAT and 5' - ACG CCG TCT TCC TCC ATC T. Primer sequences for wild-type EGFR detection were: 5' – TAC CTA TGT GCA GAG GAA TTA TGA TCT TT and 5' – CCA CTG TGT TGA GGG CAA TG. Shockingly, our results showed that only one patient (of the total of 95) was harboring EGFR vIII, therefore it was ruled out from possible resistance-drivers. Our data, alongside with another study, contradicts reports which claimed high frequency of EGFR vIII variant among HNSCC patients, and calls for clear presentation of sequencing data.

Parallel with EGFR vIII, we also screened HNSCC for EGFR R521K mutation. Among all 95 patients examined, 49 (51.6%) were found to harbor at least one allele of EGFR R521K. This enabled the possibility of a clinical relevance of the mutation on the high rate of cetuximab-resistant tumors.

Alongside with patient genotyping, we also used four well-known HNSCC cell lines - PE/CA PJ41 (mentioned as PJ41 in this report), PE/CA PJ15 (mentioned as PJ15), Cal-27 and FaDu. The cells were all negative for EGFR vIII. As for R521K polymorphism, PJ41 and Cal-27 were found homozygous wild-type, while PJ15 and FaDu were heterozygous for EGFR R521K, serving as good preclinical models for the investigation of cetuximab efficacy in EGFR wild type and EGFR R521K tumors.

First, we checked whether the cell models express different quantity of EGFR, therefore we quantified the receptor expression with immunolabeling of the cells with monoclonal antibody mAb528 and cetuximab, evaluated by flow cytometry. The results showed that no pattern was associated with R521K status (PJ41 and PJ15 cells had moderate EGFR expression, while Cal-27 and FaDu had higher receptor levels).

We started to investigate direct effects on cell proliferation and activation. We found, that up to 100 uM concentration, no toxic or antiproliferative effects were detectable on the HNSCC cells. Results are visualized on **Figure 1**.



Figure 1. Proliferation of HNSCC cells following 72-hour treatment with cetuximab.

We also examined the EGFR phosphorylation to see whether EGFR R521K causes impairment of phosphorylation inhibition. Measurement of protein levels of total and phosphorylated EGFR showed that at all common sites of EGF-dependent activation of the receptor, phosphorylation was efficiently inhibited in all cell lines, showing that the intracellular effects are alike, regardless of R521K status (**Figure 2**).

	PJ41			PJ15					Cal-27				FaDu						
control	•	•	:	· • •	• • • •	•	•	•		• • •	•	•		•	• • •	•	•		
cetuximab treated	:	•			•	•				•	•				•	•			
		EGFR pEGFR – Tyr1086		pEGFR - Tyr845			1	pEGFR – Tyr992		-	pEGFR- Tyr1045		]	pEGFR- Tyr1068		1			
	1			pEGFR- Tyr1148			1	pEGFR- Tyr1173		-	pEGFR- Ser1046/47		1	pEGFR- Ser1070					

Figure 2. EGFR and phospho-EGFR protein expression.

In-depth analysis of activation of EGFR and the possible parallel signaling driven by Hepatocyte Growth Factor Receptor (c-MET), we checked the activation of the two receptors using immunocytochemistry on PJ41 and PJ15 cells, as seen in **Figure 3**.



Figure 3. Total and phosphorylated EGFR and MET expression on HNSCC cells.

While *in vitro* tests of cetuximab-tumor cell relation showed no significant differences, we were also measuring antibody-dependent cellular cytotoxicity (ADCC) using Electric Cell-substrate Impedance Sensing system to follow up real-time cell viability of tumor cell cultures alone and in cocultures with human NK-cell derived CD16.176V.NK-92 cells. Compared to the control cocultures, cetuximab enhanced the tumor cell toxicity more effectively in EGFR wild type PJ41 and Cal-27, while it was less effective in EGFR R521K harboring PJ15 and FaDu. This difference emphasized that EGFR extracellular alteration might effectively influence cetuximab-mediated HNSCC tumor cell elimination.

To test this phenomenon, we used multiple *in vivo* xenograft models to test tumor growth and metastatic activity. Cetuximab treatment was extremely effective in EGFR wild type PJ41 and Cal-27 models, causing about complete remission of the tumors within three weeks. Interestingly, it was also effective in EGFR R521K harboring FaDu model, although it could not reduce the tumor volumes, it successfully prevented rapid tumor growth. In PJ15 xenograft model, cetuximab treatment had only modest effect, not being able to suppress tumor cell proliferation (results summarized on **Figure 4**).



Figure 4. Tumor growth and final relative tumor sizes following cetuximab therapy of HNSCC xenografts.

Besides cetuximab efficacy, we measured the antitumor effect of MET inhibitor SU11274 as an attenuator of possible parallel signaling for tumor escape in case of EGFR inhibition. MET inhibition was effective against both primary tumor growth and metastasis formation in case of the R521K PJ15 model, while, surprisingly, it had no such effect against EGFR wild type PJ41 growth or spreading. This important finding might shed light to the possible use of c-MET inhibition as future strategy against cetuximab-refractory head and neck cancers.

Our human HNSCC models were engrafted into SCID mice, a commonly used host strain lacking functional Tand B-cells. However, these mice still express activity of the cells of the innate immune response, as NK cells and macrophages. The previously performed *in vitro* ADCC experiments also emphasized the role of NK-cell activity in the mechanism of action of cetuximab. To spot the crucial cells, we performed a combined therapy experiment of cetuximab and established inhibitors of NK-cell activity and macrophage fuction. As expected, immune activity inhibition further enhanced the growth of the tested Cal-27 xenografts, while, surprisingly, these inhibitors failed to reduce the success of cetuximab against the tumors. This result was a warning signal that cetuximab antitumor effect might be more complex than a simple antibody-mediated elimination by immune cells. The combination treatment of cetuximab (CET), NK-cell inhibitor (GM1) and macrophage inhibitor (CL) could be followed on **Figure 5**.



**Figure 5.** HNSCC xenograft growth by treatment with anti-EGFR cetuximab (CET), NK-cell inhibitor (GM1) and macrophage inhibitor (CL).

While preclinical models offer a good starting point and a controllable environment to test chemotherapy efficacy, our initial motivation was to find possible mechanisms of clinical resistance or sensitivity against cetuximab in HNSCC patients. Therefore, we analysed all EGFR genotype and clinicopathological variables, correlating EGFR R521K status with EGFR intron 1 CA repeat number, immunohistological analysis of intratumoral immune cells, and therapy outcome.

The total number of patients analysed was 95. However, since the formerly used EPTF combination therapy (including additional docetaxel treatment) was discontinued, we focused on the 63 patients available receiving EPF therapy according to the EXTREME protocol. EGFR R521K genotypes of the cohort is summarized in **Table 1**. For later analyses, we excluded 4 patients confirmed to be positive for HPV infection, further investigations were performed on the cohort of 59 comparable patients.

Total	EGFR wt	EGFR R521K heterozygous	EGFR R521K homozygous			
63	29 (46%)	25 (40%)	9 (14%)			

Table 1. Number and proportion of patients with various EGFR R521K genotypes.

As EGFR intron 1 CA repeat number was proved to be associated with decreased transcriptional activity of EGFR gene, we investigated whether CA repeat number correlates with EGFR R521K geotype. As shown in **Figure 6**, there was no CA repeat (therefore protein expression suppression) difference among patient groups with various EGFR R521K status.



Figure 6. Quantification of EGFR intron 1 CA repeats among EGFR R521K groups.

As our *in vitro* and *in vivo* models raised the possibility of the importance of immune cells in cetuximab response, we performed multiple immunohistochemical analyses of HNSCC FFPE samples to quantify cells positive for CD16 (Fc $\gamma$ III – neutrophils, macrophages, NK-cells), CD68 (macrophages), or NKp46 (NK cells). The immune cell infiltration values were not different in EGFR wild type and EGFR R521K (heterozygous and homozygous polymorphic) patients, as **Figure 7** shows.



Figure 7. Immune cell infiltration of EGFR wild type and EGFR R521K HNSCC tumors

Patients	Total 59	EGFR wt 29 (49%)	EGFR R521K 30 (51%)	р
Men (n, %) Women (n, %)	44 (75%) 15 (25%)	26 (90%) 3 (10%)	18 (60%) 12 (40%)	p<0.05
Age (median, years)	57	56	58	no correlation
Overall response rate (n, %)	40 (68%)	22 (76%)	18 (60%)	no correlation
Disease contol rate (n, %)	55 (93%)	28 (97%)	27 (90%)	no correlation
Progression free survival (median, weeks)	33	35	31	no correlation
Overall survival (median, weeks)	58	63	50	no correlation

We further analysed clinical outcome of the patients, summarized in Table 2.

Table 2. Clinicopathological features of EGFR wild type and EGFR R521K HNSCC patients

In-detail analysis of the above data showed that all clinical outcome values in our cohort were showing slightly better patient response, compared to the original clinical trial results of the EXTREME protocol. It might be due to population specificity, or some refinement in diagnostic and supportive methods.

Importantly, the overall response rate, disease control rate, progression-free survival and overall survival (for the latter two, see **Figure 8.**) were not significantly different between EGFR wild type and EGFR R521K patients. These surprising results are in contrast with our preclinical findings, and, importantly, contradict to a publication previously published in the journal Cancer Research, which raises awareness of the critical thinking over published data.



Figure 8. Progression-free (left) and overall (right) survival of HNSCC patients of different EGFR R521K status.

In summary, evaluating our clinical data and comparing it with other HNSCC cohorts, our point of view is that in a clinical setup, neither EGFR R521K nor EGFR vIII alterations are suitable to determine cetuximab therapy success. Of note, preclinical results of cetuximab monotherapy are hardly comparable with the complex effect of the combination chemotherapy in clinics. Therefore, our results do not support the future introduction of EGFR extracellular alteration analysis before cetuximab treatment indication (*Manuscript submitted*). Of great importance, our finding that cetuximab-resistant HNSCCs might respond well to c-MET targeting therapy, offers a new hope to overcome therapy failure <sup>5</sup>.

## 2.2 Immune cell infiltration as predictive or prognostic factor of therapy

While EGFR R521K was not in close relation to tumor infiltrating immune cell presence, we succeeded to show that immunophenotype of the clinical HNSCC tumors were associated with survival data <sup>6</sup>. Furthermore, we published two studies on the importance of immune cell infiltration in malignant melanoma, too <sup>7,8</sup>.

### 2.3 N-glycosylation, as a potential modifying factor of protein structure and antibody binding

We successfully applied the enzymatic reaction of PNGase F in order to characterize HNSCC cells and tumor sample receptror N-glycosylation, which was suspected to alter the receptor surface, thus potentially influencing cetuximab binding. Detailed method for sample preparation from both fresh and FFPE samples was established and published <sup>9,10</sup>, however, in our clinical cohort, no characteristic changes were present for patient selection or therapy outcome prediction.

### 2.4 The role of EGFR-signaling in therapy resistance

Furthermore, we investigated different aspects of EGFR-related signaling pathways extensively. Mutations of KRAS <sup>11</sup>, BRAF <sup>12</sup>, role of MAPK and FAK signaling in Malignant mesothelioma <sup>13,14</sup>, and role of Tks4 <sup>15,16</sup> were discussed.

# 2.5 Hypoxia-driven resistance

Tumor hypoxia effects on tumor cell motility and metastatic activity were extensively analysed, including the characterization of the effects in HNSCC cell PJ15<sup>17</sup>, pointing to the effect of hypoxia in increasing migratory and metastatic activity (often associated with therapy resistance), and details of the migration process were unveiled <sup>18</sup>.

# 2.6 Targeted therapy by homing instead of antibody-binding: Cell Penetrating Peptides

Our collaborative works enabled the investigation of a different aspect of targeted therapy of tumor cells. Instead of inhibiting or immunolabeling the target moieties, peptide-drug conjugates were designed, produced and tested successfully to anchor and ingest the linked cytotoxic drug into the target cells <sup>19–23</sup>.

### 2.7 Further aspects of chemotherapy resistance

Parallel works of our lab were focused on different important aspects of therapy resistance. The selective killing of MDR tumor cells via iron depletion <sup>24</sup>, inhibition of angiogenic receptor tyrosine kinases<sup>25</sup>, and novel selective inhibitor of FLT3 receptor <sup>26</sup> were measured and published recently.

# **3 REFERENCES:**

Note: references produced in connection with the project are highlighted as **bold**.

- 1. Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. *CA: A Cancer Journal for Clinicians*. 2018;68(6):394-424. doi:10.3322/caac.21492
- 2. Vermorken JB, Mesia R, Rivera F, et al. Platinum-based chemotherapy plus cetuximab in head and neck cancer. *N Engl J Med*. 2008;359(11):1116-1127. doi:10.1056/NEJMoa0802656
- 3. Braig F, Kriegs M, Voigtlaender M, et al. Cetuximab Resistance in Head and Neck Cancer Is Mediated by EGFR-K521 Polymorphism. *Cancer Res.* 2017;77(5):1188-1199. doi:10.1158/0008-5472.CAN-16-0754
- 4. Madoz-Gúrpide J, Zazo S, Chamizo C, et al. Activation of MET pathway predicts poor outcome to cetuximab in patients with recurrent or metastatic head and neck cancer. *J Transl Med.* 2015;13:282. doi:10.1186/s12967-015-0633-7
- 5. Nelhűbel G, Cserepes M, Szabó B, et al. EGFR alterations influence the cetuximab treatment response and c-MET tyrosine-kinase inhibitor sensitivity in experimental head and neck squamous cell carcinomas. *Pathology oncology research: POR.* 2021;Accepted for publication.
- 6. Ladányi A, Kapuvári B, Papp E, et al. Local immune parameters as potential predictive markers in head and neck squamous cell carcinoma patients receiving induction chemotherapy and cetuximab. *Head Neck*. 2019;41(5):1237-1245. doi:10.1002/hed.25546
- 7. Ladányi A, Papp E, Mohos A, et al. Role of the anatomic site in the association of HLA class I antigen expression level in metastases with clinical response to ipilimumab therapy in patients with melanoma. *J Immunother Cancer*. 2020;8(1). doi:10.1136/jitc-2019-000209
- 8. Sebestyén T, Mohos A, Liszkay G, Somlai B, Gaudi I, Ladányi A. Correlation with lymphocyte infiltration, but lack of prognostic significance of MECA-79-positive high endothelial venules in primary malignant melanoma. *Melanoma Res.* 2018;28(4):304-310. doi:10.1097/CMR.00000000000457
- 9. Donczo B, Szigeti M, Ostoros G, Gacs A, Tovari J, Guttman A. N-Glycosylation analysis of formalin fixed paraffin embedded samples by capillary electrophoresis. *Electrophoresis*. 2016;37(17-18):2292-2296. doi:10.1002/elps.201500446
- 10. Donczo B, Szarka M, Tovari J, Ostoros G, Csanky E, Guttman A. Molecular glycopathology by capillary electrophoresis: Analysis of the N-glycome of formalin-fixed paraffin-embedded mouse tissue samples. *Electrophoresis*. 2017;38(12):1602-1608. doi:10.1002/elps.201600558
- 11. Kenessey I, Kói K, Horváth O, et al. KRAS-mutation status dependent effect of zoledronic acid in human non-small cell cancer preclinical models. *Oncotarget*. 2016;7(48):79503-79514. doi:10.18632/oncotarget.12806
- 12. Kenessey I, Kramer Z, István L, et al. Inhibition of epidermal growth factor receptor improves antitumor efficacy of vemurafenib in BRAF-mutant human melanoma in preclinical model. *Melanoma Res.* 2018;28(6):536-546. doi:10.1097/CMR.00000000000488
- 13. Laszlo V, Valko Z, Kovacs I, et al. Nintedanib Is Active in Malignant Pleural Mesothelioma Cell Models and Inhibits Angiogenesis and Tumor Growth In Vivo. *Clin Cancer Res.* 2018;24(15):3729-3740. doi:10.1158/1078-0432.CCR-17-1507

- 14. Laszlo V, Valko Z, Ozsvar J, et al. The FAK inhibitor BI 853520 inhibits spheroid formation and orthotopic tumor growth in malignant pleural mesothelioma. *J Mol Med (Berl)*. 2019;97(2):231-242. doi:10.1007/s00109-018-1725-7
- 15. Vas V, Kovács T, Körmendi S, et al. Significance of the Tks4 scaffold protein in bone tissue homeostasis. *Scientific Reports*. 2019;9(1):5781. doi:10.1038/s41598-019-42250-6
- 16. Vas V, Háhner T, Kudlik G, et al. Analysis of Tks4 Knockout Mice Suggests a Role for Tks4 in Adipose Tissue Homeostasis in the Context of Beigeing. *Cells*. 2019;8(8). doi:10.3390/cells8080831
- 17. Tátrai E, Bartal A, Gacs A, et al. Cell type-dependent HIF1 α-mediated effects of hypoxia on proliferation, migration and metastatic potential of human tumor cells. *Oncotarget*. 2017;8(27):44498-44510. doi:10.18632/oncotarget.17806
- 18. Keller-Pinter A, Ughy B, Domoki M, et al. The phosphomimetic mutation of syndecan-4 binds and inhibits Tiam1 modulating Rac1 activity in PDZ interaction-dependent manner. *PLoS One*. 2017;12(11):e0187094. doi:10.1371/journal.pone.0187094
- 19. Gronewold A, Horn M, Ranđelović I, et al. Characterization of a Cell-Penetrating Peptide with Potential Anticancer Activity. *ChemMedChem.* 2017;12(1):42-49. doi:10.1002/cmdc.201600498
- 20. Tripodi AAP, Ranđelović I, Biri-Kovács B, Szeder B, Mező G, Tóvári J. In Vivo Tumor Growth Inhibition and Antiangiogenic Effect of Cyclic NGR Peptide-Daunorubicin Conjugates Developed for Targeted Drug Delivery. *Pathol Oncol Res.* 2020;26(3):1879-1892. doi:10.1007/s12253-019-00773-3
- 21. Bánóczi Z, Keglevich A, Szabó I, et al. The effect of conjugation on antitumor activity of vindoline derivatives with octaarginine, a cell-penetrating peptide. *J Pept Sci.* 2018;24(10):e3118. doi:10.1002/psc.3118
- 22. Kapuvári B, Hegedüs R, Schulcz Á, et al. Improved in vivo antitumor effect of a daunorubicin - GnRH-III bioconjugate modified by apoptosis inducing agent butyric acid on colorectal carcinoma bearing mice. *Invest New Drugs*. 2016;34(4):416-423. doi:10.1007/s10637-016-0354-7
- 23. Ranđelović I, Schuster S, Kapuvári B, et al. Improved In Vivo Anti-Tumor and Anti-Metastatic Effect of GnRH-III-Daunorubicin Analogs on Colorectal and Breast Carcinoma Bearing Mice. *Int J Mol Sci.* 2019;20(19). doi:10.3390/ijms20194763
- 24. Cserepes M, Türk D, Tóth S, et al. Unshielding Multidrug Resistant Cancer through Selective Iron Depletion of P-Glycoprotein-Expressing Cells. *Cancer Res.* 2020;80(4):663-674. doi:10.1158/0008-5472.CAN-19-1407
- 25. Torok S, Rezeli M, Kelemen O, et al. Limited Tumor Tissue Drug Penetration Contributes to Primary Resistance against Angiogenesis Inhibitors. *Theranostics*. 2017;7(2):400-412. doi:10.7150/thno.16767
- 26. Baska F, Sipos A, Őrfi Z, et al. Discovery and development of extreme selective inhibitors of the ITD and D835Y mutant FLT3 kinases. *Eur J Med Chem.* 2019;184:111710. doi:10.1016/j.ejmech.2019.111710